Author's response to reviews

Title: Testing the association between psychosocial job strain and adverse birth outcomes. Design and methods.

Authors:

Ann D Larsen (adl@nrcwe.dk)
Harald Hannerz (hha@nrcwe.dk)
Carsten Obel (co@alm.au.dk)
Ane M Thulstrup (anethuls@rm.dk)
Jens P Bonde (jbbon0004@bbh.regionh.dk)
Karin S Hougaard (ksh@nrcwe.dk)

Version: 5 Date: 15 April 2011

Author's response to reviews:

Changes made to the formatted version of the paper:
- P. 3, l. 10: "To our knowledge eight studies" has been replaced by “Several”.
- P. 8, l. 16: A comma (,) has been inserted between “SGA” and “i.e.”.
- P. 9, l. 6: “count” has been replaced by “counts”.
- P. 11, l. 3: Inserted “an” between “than” and “adverse”